Author's response to reviews

Title: Lymph nodal Fine Needle Cytology in the Staging and Follow-up of Cutaneous Lymphomas

Authors:

Elena Vigliar (elena@vigliar.it)
Immacolata Cozzolino (coimma@yahoo.it)
Anna Lucia Peluso (peluso@dbbm.unina.it)
Laura Virginia Sosa Fernandez (laurasosa@hotmail.it)
Claudio Bellevicine (claudiobellevicine@gmail.com)
Giancarlo Troncone (giancarlo.troncone@unina.it)
Marco Picardi (marco.picardi@unina.it)
Fabrizio Pane (fabrizio.pane@unina.it)
Carnine Selleri (cselleri@unisa.it)
Pio Zeppa (pzeppa@unisa.it)

Version: 4 Date: 13 November 2013

Author's response to reviews:

1) Ethics statement
The approval of the study obtained by the Ethic Committee of the Istituto Nazionale Tumori Fondazione “G. Pascale”, of Naples, Italy, and the informed consent for the FNC performing, the diagnostic procedures and the scientific use of biological material, obtained from all the patients are now mentioned in materials and methods.

2) Please remove or modify table 1 so that no data that could lead to patient identification is shown, for example by presenting the data as ranges or percentages.

In Table 1 in the present version, the patient identification is impossible.